Author's response to reviews

Title: Antecedents of teenage pregnancy from a 14-year follow-up study using data linkage

Authors:

Jennifer Gaudie (shenny@iinet.net.au)
Francis Mitrou (francism@ichr.uwa.edu.au)
David Lawrence (D.Lawrence@curtin.edu.au)
Fiona J Stanley (fiona@ichr.uwa.edu.au)
Sven R Silburn (sven@ichr.uwa.edu.au)
Stephen R Zubrick (steve@ichr.uwa.edu.au)

Version: 4 Date: 27 January 2010

Author's response to reviews:

Minor changes have been made at the proofing stage to correct a couple of errors, a double space between words, some inconsistent formatting in the reference list